MCC Operations Status Meeting

January 28, 2008
This afternoon at the 4:15 pm meeting:

Everyone:

Paul Emma’s goal by the end of this week is to have the LCSHOME alarms “green”

We are very close, the subsystems that are showing alarms this evening are BPMs, Magnets, Events, Feedback, LLRF and Toroids.

Computer Systems:  Jingchen Zhou

1. The physics e-log virtual print queue font size is very small, much smaller when printing to the paper.

2. The print queue names on the opi’s should match the 

3. The print queues on lcls-prod02 are not setup so that the mcc users can print to the printers in mcc and the e-logs (mcc and physics).  One use case is when you bring up firefox on lcls-prod02  from an mcc opi  the operator cannot currently select a printer.
4. Sunray CUD displays go away when your reboot and logout of mcc opi’s. 

5. Try to add more processes watchdogs that set alarms 
Laser:  Sheng Peng and Zen Szalata

1. The camera ORT2 shows a blank vertical strip on the right side of the display.
Model: Greg White

1. BPMS LI24 201 from this morning came up again. 

Greg checked with Rick Iverson to see if he can add this bpm unit to the model.

Toroids: Sheng Peng, Zen Szalata, Arturo Alarcon, Mike Cecere

1. Toroid comparator faults causing MPS trips. Operators have bypassed.

This morning at the 8:00 am meeting:
Everyone:

Paul Emma’s goal by the end of this week is to have the LCSHOME display be “green”, no alarms.

Aida: Greg White

Aida server went down. Aida has a dependency on an SCCS Serve that went down. It is not clear if this was a scheduled down.

BIC:  Mike Lazmo
1. Artimis # 82548 – The software gateway lost connection to bic06pep00 pv’s BICL:DCCT:SUMY -> LB60:DCCT:SUMY and BICH:DCCT:SUMY -> HB60:DCCT:SUMY. Rebooting the BIC cleared up problem.

2. All pv’s on the ioc bic06pep00 go white. Previously a problem was reported that there were not enough network buffers available on the ioc, which caused the BIC pv’s to go white. 

Computer Systems: Jingchen Zhou

1 Artimis #82534, #82538 – The LCLS Klystron CUD froze after which the users were not able to restart this CUD. Zoe said that there are instructions for restarting this CUD although software should talk to ops because the instructions need some changes. This CUD appears to stop working quite often.

Magnets: Kristi Luchini

1. LEM unable to TRIM LM24 magnets.

Model:  Greg White 

1. Atrimis #82546 – BPMS LI24.201 appears to be missing from the model. See Jim Turner or Paul Emma as well as plots in the physics log from J. Turner dated 1/27/2008 12:25pm. Orbit fitting possibly boundary problem between li23 and li24, offset by 1 bpm unit
MPS: Ken Underwood

1. Artimis #82542 – MPG hangs due to full mailbox. This occurred about a month ago and a fix determined although it had not been put in place yet. Ken Underwood resolved the problem on Sunday.

SCORE: Debbie Rogind
Load of BXKIK BCON doesn’t  work and there is no matching unit on LM25 in the SLC database Initially there was an LGPS.LI25.3 unit used for BXKIK but it was later decide that the KICK primary would be used. This change meant that BXKIK is an EPICS only pv for magnets and there is no equivalent SLC unit number. To load BCON from SCORE use  KICK:LI25:344:BCON.B rather than KICK:LI25:344:BCON.
Toroid:  Mike Cecere and Arturo Alarcon

1.  Artimis #82547 – Toroid comparator showed faults when there was no beam present, maybe pedestal problem. Missing toroid comparator alarm “red” on the LCLSHOME page and cannot find values on an EPICS display. Symptom of toroid comparator problem may have been that taking backgrounds on OTR2 did not work - gives all bad data

.

